Author's response to reviews

Title: Factors associated with the rejection of active euthanasia: A survey among the general public in Austria.

Authors:

Willibald J Stronegger (willibald.stronegger@medunigraz.at)
Nathalie T Burkert (nathalie.burkert@medunigraz.at)
Franziska Grossschäd (franziska.grossschaedl@medunigraz.at)
Wolfgang Freidl (wolfgang.freidl@medunigraz.at)

Version: 2 Date: 26 November 2012

Author's response to reviews: see over
Dear Editors,

We would like to resubmit the enclosed paper about “Factors associated with the rejection of active euthanasia: A survey among the general public in Austria” for consideration for publication in the BMC Medical Ethics.

The Ethics Committee of the Medical University of Graz waives the necessity of an ethical approval in case of studies based on questionnaire data from population surveys.

We therefore included the statement

"The Ethics Committee of the Medical University of Graz waived the necessity of an ethical approval."

into the first paragraph of the methods section (page 6).

Kind regards,

(Willibald J. Stronegger, PhD)

Corresponding author:

Ao. Univ.-Prof. Dr. Willibald J. Stronegger
Medical University of Graz, Austria
Institute of Social Medicine and Epidemiology
Medical University of Graz, 8010 Graz